Author's response to reviews

Title: Immunologically reactive M. leprae antigens with relevance to diagnosis and vaccine development.

Authors:

Lucas H. Sampaio (lucashfs@gmail.com)
Mariane M. Stefani (mstefani@iptsp.ufg.br)
Regiane M. Oliveira (remorillas@gmail.com)
Ana Lucia . M. Sousa (anamaroclo@gmail.com)
Greg C. Ireton (gireton@idri.org)
Steven G. Reed (sreed@idri.org)
Malcolm S. Duthie (mduthie@idri.org)

Version: 3 Date: 30 November 2010

Author's response to reviews: see over
TO BMC INFECTIOUS DISEASES

Dear Editor

We are resubmitting to BMC Infectious Diseases the revised form of our manuscript entitled: “Immunologically reactive *Mycobacterium leprae* antigens with relevance to diagnosis and vaccine development”.

We thank both reviewers for their comments and queries understanding the extensive nature of the analyses contained in this manuscript. We have prepared a modified version highlighting the changes made according to reviewers’ request. Another version of the modified manuscript without highlights is attached. We have also prepared a point by point rebuttal to both reviewers.

Regards,
Professor Mariane Martins de Araújo Stefani-
Tropical Pathology and Public Health Institute
Federal University of Goias/ Brazil